\begin{thebibliography}{26}
\expandafter\ifx\csname natexlab\endcsname\relax\def\natexlab#1{#1}\fi

\bibitem[Andersen(1996)]{Andersen}
K.~D. Andersen.
\newblock A modifies schur complement method for handling dense columns in
  interior point methods for linear programming.
\newblock {\em ACM Transactions on Mathematical Software}, 22\penalty0
  (3):\penalty0 348--356, 1996.

\bibitem[Bennet(1965)]{Bennet}
J.~M. Bennet.
\newblock Triangular factors of modified matrices.
\newblock {\em Numerisches Mathematik}, 7:\penalty0 217--221, 1965.

\bibitem[Blake and Merz(1998)]{BM98}
C.~L. Blake and C.~J Merz.
\newblock {UCI} repository of machine learning databases, 1998.

\bibitem[Boser et~al.(1992)Boser, Guyon, and Vapnik]{BGV92}
B.~Boser, I.~Guyon, and V.~N. Vapnik.
\newblock A training algorithm for optimal margin classifiers.
\newblock In D.~Haussler, editor, {\em Proceedings of the 5th Annual ACM
  Workshop on Computational Learning Theory}, pages 144--152. ACM Press, 1992.

\bibitem[Choi et~al.(1990)Choi, Monma, and Shanno]{ChoiMonmaShanno}
I.~C. Choi, C.~L. Monma, and D.~F. Shanno.
\newblock Further development of primal-dual interior point methods.
\newblock {\em ORSA J. on Computing}, 2\penalty0 (4):\penalty0 304--311, 1990.

\bibitem[Fine et~al.(2001)Fine, Navr\'atil, and Gopinath]{FNG01a}
S.~Fine, J.~Navr\'atil, and R.~A. Gopinath.
\newblock A hubrid gmm/svm approach to speaker identification.
\newblock In {\em Proc. of the International Conference on Acoustics, Speech,
  and Signal Processing (ICASSP)}, 2001.

\bibitem[Fine and Scheinberg(2001)]{FS2}
S.~Fine and K.~Scheinberg.
\newblock Efficient application of interior point methods for quadratic
  problems arising in support vector machines using low-rank kernel
  representation.
\newblock Submitted to {\sl Mathematical Programming}, 2001.

\bibitem[Fletcher and Powell(1974)]{FP}
R.~Fletcher and M.~J.~D. Powell.
\newblock On the modification of $ldl\t$ factorization.
\newblock {\em Mathematics of Computation}, 28\penalty0 (128):\penalty0
  1067--1087, 1974.

\bibitem[Friess et~al.(1998)Friess, Cristianini, and Campbell]{FCC98}
T.~T. Friess, N.~Cristianini, and C.~Campbell.
\newblock The kernel-adaraton algorithm: A fast simple learning procedure for
  support vector machines.
\newblock In {\em Proceedings of the 15th International Conference on Machine
  Learning}, pages 188--196. Morgan Kaufman, 1998.

\bibitem[Gill et~al.(1975)Gill, Murray, and Saunders]{GMS}
Ph.~E. Gill, W.~Murray, and M.~A. Saunders.
\newblock Methods for computing and modifying the ldv factors of a matrix.
\newblock {\em Mathematics of Computation}, 29:\penalty0 1051--10, 1975.

\bibitem[Goldfarb and Scheinberg(1999)]{GS1}
D.~Goldfarb and K.~Scheinberg.
\newblock A product-form cholesky factorization method for handling dense
  columns in interior point methods for linear programming.
\newblock Submitted, 1999.

\bibitem[Golub and Van~Loan(1996)]{GolubVanLoan}
G.~H. Golub and Ch.~F. Van~Loan.
\newblock {\em Matrix Computations}.
\newblock The Johns Hopkins University Press, Baltimore and London, 3 edition,
  1996.

\bibitem[Jaakkola and Haussler(1999)]{JH99}
T.~S. Jaakkola and D.~Haussler.
\newblock Exploiting generative models in discriminative classifiers.
\newblock In M.~S. Kearns, S.~A. Solla, and D.~A. Cohn, editors, {\em Advances
  in Neural Information Processing Systems}, volume~11. MIT Press, 1999.

\bibitem[Joachims(1999)]{Joachims99}
T.~Joachims.
\newblock Making large-scale support vector machine learning practical.
\newblock In B.~Sch\"{o}lkopf, C.~C. Burges, and A.~J. Smola, editors, {\em
  Advances in Kernel Methods}, chapter~12, pages 169--184. MIT Press, 1999.

\bibitem[Marxen(1989)]{Marxen}
A.~Marxen.
\newblock Primal barrier methods for linear programming.
\newblock Technical report, Dept. of Operations Research, Stanford University,
  Stanford, CA, 1989.

\bibitem[Mehrotra(1992)]{Mehrotra}
S.~Mehrotra.
\newblock On implementation of a primal-dual interior point method.
\newblock {\em SIAM J. on Optimization}, 2\penalty0 (4):\penalty0 575--601,
  1992.

\bibitem[Oliver et~al.(2000)Oliver, Sch\"{o}lkopf, and Smola]{OSS00}
N.~Oliver, B.~Sch\"{o}lkopf, and A.~J. Smola.
\newblock Natural regularization from generative models.
\newblock In A.~J. Smola, B.~Sch\"{o}lkopf, P.~L. Bartlett, and D.~Schuurmans,
  editors, {\em Advances in Large Margin Classifiers}, chapter~4, pages 51--60.
  MIT Press, 2000.

\bibitem[Osuna et~al.(1997)Osuna, Freund, and Girosi]{OFG97}
E.~Osuna, R.~Freund, and F.~Girosi.
\newblock An improved training algorithm for support vector machines.
\newblock In {\em Proceedings of the IEEE Neural Networks for signal Processing
  VII Workshop}, pages 276--285. IEEE, 1997.

\bibitem[Platt(1999)]{Platt99}
J.~C. Platt.
\newblock Fast trining support vector machines using sequential mininal
  optimization.
\newblock In B.~Sch\"{o}lkopf, C.~C. Burges, and A.~J. Smola, editors, {\em
  Advances in Kernel Methods}, chapter~12, pages 185--208. MIT Press, 1999.

\bibitem[Scheinberg and Wright(2000)]{SW}
K.~Scheinberg and S.~Wright.
\newblock A note on modified cholesky and schur complement in interior point
  methods for linear programming.
\newblock Manuscript, 2000.

\bibitem[Smola and Sch\"{o}lkopf(2000)]{SS00}
A.~J. Smola and B.~Sch\"{o}lkopf.
\newblock Sparse greedy matrix approximation for machine learning.
\newblock In {\em Proceedings of the 17th International Conference on Machine
  Learning}, pages 911--918, Stanford University, CA, 2000. Morgan Kaufmann
  Publishers.

\bibitem[Vapnik(1995)]{Vap95}
V.~N. Vapnik.
\newblock {\em The Nature of Statistical Learning Theory}.
\newblock Springer-Verlag, 1995.

\bibitem[Williams and Seeger(2001)]{WS01}
C.~Williams and M.~Seeger.
\newblock Using the nystr{\"o}m method to speed up kernel machines.
\newblock In Todd~K. Leen, Thomas~G. Dietterich, and Volker Tresp, editors,
  {\em Advances in Neural Information Processing Systems 13}, pages 682--688.
  MIT Press, 2001.

\bibitem[Williamson et~al.(1998)Williamson, Smola, and Sch\"{o}lkopf]{WSS98}
R.~C. Williamson, A.~J. Smola, and B.~Sch\"{o}lkopf.
\newblock Generalization performance of regularization networks and support
  vector machines via entropy numbers of compact operators.
\newblock NeuroCOLT NC-TR-98-019, Royal Holloway College, University of London,
  UK, 1998.

\bibitem[Wright(1996)]{Wright2}
S.~Wright.
\newblock Modified cholesky factorizations in interior point algorithms for
  linear programming.
\newblock Preprint anl/mcs-p600-0596, Argonne National Laboratory, Argonne, IL,
  1996.

\bibitem[Wright(1997)]{Wright}
S.~Wright.
\newblock {\em Primal-Dual Interior Point Methods}.
\newblock SIAM, Philadelphia, 1997.

\end{thebibliography}


\section{Low-rank updates}
\label{sec:Low-rank_updates}
There are two methods which efficiently solve the system $(D+VV\t)u=w$
by taking into consideration the special form of the matrix
of the system (i.e., the fact that it is a diagonal matrix plus
a low-rank symmetric positive semidefinite matrix).
Both of these
methods require $O(nk)$ storage and $O(nk^2)$ arithmetic operations, 
where $k$ is the
number of columns in $V$. Thus, if $k<<n$ then the two methods provide 
significant savings in terms of workload and storage.

The first method is somewhat cheaper than the second. Its workload
is about half of that of the second method and the storage requirement is
one third of the storage of the second method. 
On the other hand, the first method may experience numerical instability.

\subsection{Sherman-Morrison-Woodbury update}

We  consider a  classical  method for a low-rank update of 
an inverse of a matrix. This update is usually called the 
Sherman-Morrison-Woodbury (SMW) update or formula:
$$
(D+VV\t)^{-1}=D^{-1}-D^{-1}V(I+V\t D^{-1}V)^{-1}V\t D^{-1}.
$$
Since we would like to avoid storing the matrix $D+VV\t$, we also would like
to avoid storing its inverse. We can apply the above formula to
solve the system $(D+VV\t)u=w$ as follows: 
$$
u=D^{-1}w-D^{-1}V(I+V\t D^{-1}V)^{-1}V\t D^{-1}w. 
$$
Hence, compute 
\begin{eqnarray}
&& z:=D^{-1}w \nonumber\\
&& t:\ (I+V\t D^{-1}V)t=V\t z\nonumber \\
&& u:=z-D^{-1}Vt. \nonumber 
\end{eqnarray}

The system in the second step is $k \times k$ symmetric positive
definite and can be solved by computing  Cholesky factorization
of $(I+V\t D^{-1}V)$. The work required to compute such factorization
is $O(k^3)$, which is small if $k$ is small with respect to $n$.
Computing the matrix of that system takes $nk^2/2+O(nk)$ multiplications
(and the same number of additions). The overall work, in terms
of the number of multiplications, required to
solve $(D+VV\t)u=w$ is $nk^2/2$ plus smaller order terms.

The Sherman-Morrison-Woodbury has been used widely in
the context of interior point methods for linear programming,
\citep{ChoiMonmaShanno, Marxen}.
In that context, however, the method often runs into numerical 
difficulties. There are few modifications of the method
 that have been proposed to deal with the numerical instability
\citep{Andersen,SW}. However, these modifications do not
always help, and they increase the computational cost.

It is not easy to point out  rigorously  the situations
for which the SMW
formula is numerically unstable.
The intuition suggests that it happens when matrix $D+VV\t$ or
just $D$ is not well conditioned. 
Consider the following small example:
\begin{example}\label{exmp:SMW}
Let
$$
D=\left [\begin{array}{cc} \epsilon^2  &  \\  & 1 \end{array} \right ]
\quad {\rm and} \quad V=\left [\begin{array}{r}1\\-1\end{array}\right ].
$$
Then assuming that $\epsilon ^2$ is smaller than the relative machine 
precision - i.e., $1
\pm\epsilon^2$ is computed as $1$ - we have that 

\begin{equation}
D +VV\t\approx \left [\begin{array}{rr}1 &  -1\\  -1& 2\end{array}\right ]. \nonumber
\end{equation}
Now consider solving $(D+VV\t)u=w$, where $w=\left ( \begin{array}{c} w_1 \\ w_2 \end{array} \right )$ and $w_1$ and $w_2$ are $\Theta(1)$. Clearly
\begin{equation}
u\approx \left( \begin{array}{c}{2w_1+w_2}
\\w_1+w_2
\end{array}\right ). \label{solut}
\end{equation}

Let us apply the Sherman-Morrison-Woodbury  formula to this example.  
$$
z=\left (\begin{array}{c}{w_1}\over{\epsilon^2} \\ w_2
\end{array}\right )
$$
$$
(V\t D^{-1} V+I)t=
(2+{ {1}\over{\epsilon^2}})t \quad {\rm and} \quad V\t z={{w_1}\over{\epsilon^2}}-w_2.
$$
Because of round-off errors $V\t D^{-1} V+I$ is computed as 
$1/\epsilon^2$ and $V\t z$ as  $w_1/\epsilon^2$.
Hence $t=w_1$,
$$
u=z-D^{-1}Vt=\left (\begin{array}{c}
{{w_1}\over{\epsilon^2}}-{{w_1}\over{\epsilon^2}} \\ w_2+w_1 
\end{array}\right )=\left (\begin{array}{c}
0\\ w_2 +
w_1\end{array}\right ).
$$
Clearly, the computed solution is not even close to the approximately 
correct solution (\ref{solut}).
%\vskip0.5cm
%
\end{example}
A tuation similar to the above example happens often in linear programming.
For our QP problems such a situation is less natural, but still possible.
The diagonal elements of $D$ are $s_i/x_i+\xi _i/(c-x_i)$. By complementarity,
as $\mu$ approaches zero, either $s_i\to 0$ and $x_i \to x_i^*>0$ or
$x_i\to 0$ and $s_i \to s_i^*>0$ (analogously, either $\xi_i\to 0$ and $x_i \to x_i^* < c$ or
$x_i\to c$ and $\xi_i \to \xi_i^*>0$). Moreover, whenever a variable
converges to zero, it converges with the same rate as the parameter $\mu$.
Thus, whenever $s_i$ and $\xi_i$ both converge to zero (which happens
for the points that are in the active optimal set), then $D_i$ converges
to zero with the same rate as $\mu$. Otherwise, $D_i$ converges to
infinity with the same rate as $1/\mu$. If $\mu=10^{-8}$, then
some elements of $D$ are of the order $10^{-8}$ and the others are of the
order $10^8$. If we scale $D$ and $Q$ by $10^{-8}$ then $D$ looks 
similar to the diagonal matrix in the example; i.e., some of the elements are
of the order of $1$ and some are of the order of $10^{-16}$
(which often is the order of  the machine precisions).
To have matrix $Q$ and $V$ consistent with the example, we need the 
elements of scaled matrix $10^{-8}Q$ to be of the order of $1$.
This will happen if the elements of $V$, e.g.  the data points, 
 are of the order $10^4$. 
In section \ref{sec:Experiments}
we briefly discuss some examples of practical SVM problems 
for which our implementation
of the interior point method with the SMW update failed to converge.



In the next subsection we present an alternative method, which avoids many
of the numerical difficulties of the SMW update.



\subsection{Product-Form Cholesky Factorization}\label{pfcf}
An alternative efficient method for solving the
system $(D+VV\t)u=w$ is based on a low rank
update of the Cholesky factorization of the matrix,
rather than using low rank updates to the inverse of the matrix. 
This is the reason for good numerical properties of this approach.
We will use a, so called, product form Cholesky factorization
of $D+VV\t$.

Here a Cholesky factorization of a symmetric positive semidefinite matrix 
$M\in \R^{n\times n}$ is defined as $M=L\Lambda L\t$, where $L$ is a 
lower triangular matrix with
ones on the diagonal, and $\Lambda$ is a nonnegative diagonal matrix.
A product form Cholesky factorization, in general, is of the form 
$\lt ^1 \lt ^2 \ldots \lt ^k \dt ^k (\lt^k)\t \ldots (\lt^2)\t (\lt^1)\t$, where
each $\tilde L^i$ is a lower triangular matrix and $\dt^k$ is a nonnegative
diagonal matrix. 

To solve a system of linear equations $Mu=w$, one has to solve the following
sequence of linear systems: $\tilde L^1 u^1=w$, 
$\tilde L^2 u^2=u^1$, $\ldots$ , $\tilde L^k u^k=u^{k-1}$, $(\tilde L^k)\t 
u^{k+1}=(\Lambda^k)^{-1}u^{k}$, $\ldots$ , $(\tilde L^1)\t u=u^{2k-1}$. 
Each such system has a lower triangular or an upper triangular matrix,
and, in a  general dense case, solving each such system takes $O(n^2)$ 
operations. However, if each $\tilde L^i$ has a special structure
(such as we will see below) then solving each system takes only $O(n)$ 
operations.

Let us assume that we have a Cholesky factorization of a matrix $M$:
$M=L\Lambda L\t$. We would like to obtain a product form factorization
of $M+vv\t$, where $v\in \R ^n$ is a vector (hence, $vv\t$ is a rank-one
$n\times n$ matrix):
$$
L\Lambda L\t +vv\t= L(\Lambda +pp\t)L\t=
L\lt \tilde \Lambda \tlt L \t,
$$
where $p$ is the solution of the equations $Lp=v$
and $\lt \tilde \Lambda \tlt$ is the Cholesky factorization of 
$\Lambda +pp\t$. 

$\tilde L$ has a special form
\begin{equation}
\lt=\left [ \begin{array}{cccccc} 1 & & & & & \\
            p_2\beta _1 & 1 & & & & \\
            p_3\beta _1 &  p_3 \beta_2 & 1 & & &\\
            \vdots & \vdots & \vdots & \ddots & & \\
            p_{n-1}\beta _1 & p_{n-1}\beta_2 & p_{n-2}\beta _3& \cdots & 1 &\\
            p_n\beta _1 & p_n \beta _2 & p_n \beta _3 & \cdots & p_n\beta_{n-1}
            & 1 \end{array} \right ], \label{ltild}
\end{equation}
where $\beta\in \R ^n$ and $\tilde \Lambda={\rm diag}\{\tilde\lambda_1, \ldots,
\tilde\lambda _n\}$ can be computed from the following recurrence relations:
\begin{eqnarray}
&& t_0:=1, \nonumber\\
&& {\rm for \ } j=1,2,\ldots,n \nonumber \\
&& \hskip0.5cm t_j:=t_{j-1} +p_j^2 /\lambda_j , \nonumber \\
&& \hskip0.5cm \tilde \lambda_j :=\lambda_j t_j/t_{j-1}, \nonumber\\
&& \hskip0.5cm \beta _j :=p_j/(\lambda_jt_j). \nonumber 
\end{eqnarray}
Various versions of this method 
for updating Cholesky factorization 
by a rank-one term were  proposed by 
\citet{Bennet}, \citet{FP} and  \citet{GMS}.

If we allow some  $\lambda _j$ to be zero,  then it is easy
to verify that the above recurrence relations 
become:
\begin{figure}[htb]
\centerline{
\fbox{\begin{minipage}{29pc}
\begin{eqnarray}
&& t_0:=1, \nonumber\\
&& {\bf for \ } j=1,2,\ldots,n \nonumber \\
&& \hskip0.5cm {\bf if \ } t_{j-1}\neq \infty \nonumber \\
&& \hskip1cm {\bf if \ } \lambda_j\neq 0 \nonumber \\
&& \hskip1.5cm t_j:=t_{j-1} +p_j^2 /\lambda_j , \nonumber \\
&& \hskip1.5cm \tilde \lambda_j :=\lambda_j t_j/t_{j-1}, \nonumber\\
&& \hskip1.5cm \beta _j :=p_j/(\lambda_jt_j). \nonumber \\
&& \hskip1cm{\bf else \ }  \nonumber \\
&& \hskip1.5cm {\bf if\ } p_j\neq 0\nonumber \\
&& \hskip2cm t_j:=\infty \nonumber \\
&& \hskip2cm \tilde \lambda_j :=p_j^2/t_{j-1}, \nonumber \\
&& \hskip2cm \beta _j :=1/p_j. \nonumber \\
&& \hskip1.5cm{\bf else } \nonumber \\
&& \hskip2cm t _j :=t_{j-1} \nonumber \\
&& \hskip2cm \tilde \lambda _j :=0 \nonumber \\
&& \hskip2cm \beta _j :=0 \nonumber \\
&& \hskip0.5cm{ \bf else \ } \nonumber \\
&& \hskip1cm t_j:=\infty \nonumber \\
&& \hskip1cm \tilde \lambda_j :=\lambda _j, \nonumber\\
&& \hskip1cm \beta _j :=0. \nonumber
\end{eqnarray}
\end{minipage}}
}
\caption{Rank-One update for Product-Form Cholesky Factorization}
\label{fig:RankOnePFCF}
\end{figure}

It is often important for the purpose of numerical stability
to consider $\lambda_j$ as equal to zero, even if it is not
precisely zero, but is very small (relatively to the other 
elements of $\Lambda$). The above recursions can be used
(with slight modifications) in the approximate sense as well.

Now assume that we  have a factorization $L\Lambda L\t$ and we 
want to obtain a product form factorization
of $L\Lambda L\t+v^1(v^1)\t+v^2(v^2)\t+\ldots+v^k(v^k)\t$.
By repeating the above procedure $k$ times, once for each rank-one term 
$v^i(v^i)\t$, we can obtain the product from Cholesky factorization 
$L \lt ^1 \lt ^2 \ldots, \lt ^k \dt ^k (\lt^k)\t \ldots  (\lt^2)\t (\lt^1)\t  
\tl$
from the following:

\vskip0.5cm

\noindent {\bf Procedure 1.}
\begin{description}

\item[1.] Set $\dt ^0:=\Lambda$,

\item[2.] For $i=1,\ldots, k$

\begin{description}
\item[(i)] Solve $Lq^{0,i}=v^i$ for $q^{0,i}$;
\item[(ii)] For $j=1,\ldots, i-1$, solve $\lt ^j q^{j,i}=q^{j-1,i}$ for $q^{j,i}$;
\item[(iii)] Set $p:=q^{i-1,i}$; compute $\beta$ and $\dt $ by recurrence
relations in Fig. \ref{fig:RankOnePFCF};
\item[(iv)] Store $p^i:=p$ and $\beta ^i:=\beta$; replace $\dt ^{i-1}$ by $\dt ^i$.
\end{description}
\end{description}

\vskip0.5cm

In the case of our QP problems, we need to compute product form
Cholesky factorization of a matrix $M=D + VV\t$, where $D$ is a diagonal matrix
in $\R^{n\times n}$, $V$ is a rectangular matrix in $\R^{n\times k}$ 
(assuming $k<<n$), 
and $VV\t=\sum_{i=1}^k v^i(v^i)\t$, where $v^i$ is the $i$-th column
of $V$. Thus, we can apply the above Procedure 1 by setting
initial $\Lambda$ to equal $D$ and  $L$ to $I_n$, an $n\times n$ identity 
matrix.

To store information about the  product form Cholesky factorization of 
$(D+VV\t)$ we need to store the matrix $V$, the diagonal elements of 
$\Lambda^k$, $k$ vectors
$p^i$ and $k$ vectors $\beta^i$, $i=1,\ldots k$.
Overall, our memory requirement for the factorization is $3kn$ plus smaller
order terms.

To compute the vectors $p^i$ and $\beta^i$, $i=1, \ldots, k$, we need to solve
$(k-1)(k-2)/2$ systems of the form $\tilde Lq=r$ (by $\tilde L$, 
without an index, we mean any matrix of the form of Eq. (\ref{ltild})).
Due to the special structure of $\tilde L$ such solution can be
obtained by the following procedure:

\vskip0.5cm
\noindent {\bf Procedure 2.}

Given $p$, $\beta $ and $r$;

\begin{description}
\item[1.] Set $q:=r$ and  $\sigma:=q_1\beta _1$;
\item[2.] For $j=2,\ldots, n$
\begin{description} 
 \item Set $q_j:=q_j-p_j\sigma$,
 \item  ${\sigma}:={\sigma}+q_j\beta _j$
\end{description}
\end{description}

\vskip0.5cm

This procedure requires $2n$ multiplications 
(and the same number of additions). Recursions
in Fig. \ref{fig:RankOnePFCF}  require $O(n)$ operations for each $i=1,\ldots, k$.
Hence, overall, the computation of the product from Cholesky factorization
of $D+VV\t$ requires $k^2n$ ( plus smaller order term) multiplications.

Clearly, solving the system of equations of the form $\tilde L\t q=r$,
and performing matrix multiplication $q=\tilde L r$ or $q=\tilde L\t r$ 
each can be done by a procedure which takes into account the special structure
of $\tilde L$, similarly to Procedure 2, and  
requires $O(n)$ operations. Thus, once the factorization of $D+VV\t$ is 
computed,
solving $(D+VV\t)u=w$  requires only $O(kn)$ operations. The overall 
complexity in terms of the number of multiplications is then 
$k^2n$ plus smaller order term. This is about twice as much as the complexity
of solving this system using the SMW update. However, if $k<<n$ this
complexity is still a very significant improvement over $O(n^3)$ 
operations required to  solve $(D+VV\t)u=w$ directly.

The product form Cholesky factorization method is numerically more stable
than the SMW method both in theory and in practice. First, it is easy to
check that it produces a correct result when applied to the
small example in the previous subsection.
Second, in Section \ref{sec:Experiments}
we show some examples of QP for which the implementation of 
an interior point method using the product form Cholesky factorization 
works well, while the implementation of the same interior point method
with SMW update fails numerically. 
Finally, in \citet{GS1} it is
shown that the product form Cholesky factorization is numerically 
stable when used in the context of interior point method for linear programming
(under the assumption that the initial data is well conditioned).
This result is supported by extensive computational 
evidence. The case of QP for SVM is somewhat different from
the case of linear programming. The proof of numerical stability 
does not apply directly, and extending it is tedious and beyond 
interest of this paper;
however, similar intuition applies. Computational experiments show that the product form 
Cholesky factorization method avoids numerical difficulties even when the initial data
is not well conditioned.









